Author's response to reviews

Title: Frequent detection of Bocavirus DNA in German children with respiratory tract infections

Authors:

Benedikt B Weissbrich (weissbrich@vim.uni-wuerzburg.de)
Florian F Neske (florian.neske@gmx.de)
Jorg J Schubert (schubert@vim.uni-wuerzburg.de)
Franz F Tollmann (tollmann@vim.uni-wuerzburg.de)
Katharina K Blath (blathk@web.de)
Kerstin K Blessing (blessing_k@kinderklinik.uni-wuerzburg.de)
Hans Wolfgang HW Kreth (kreth@mail.uni-wuerzburg.de)

Version: 3 Date: 9 July 2006

Author's response to reviews:

Dear Editorial Team,

In the revised version of the manuscript, the requested format changes (replacement of Fig. by Figure) have been made.

With best regards,
Benedikt Weissbrich